

The Price of Purity: Willingness to Pay for Air and Water Purification Technologies in Rajasthan, India

Alexandra K. Shannon¹ · Faraz Usmani² · Subhrendu K. Pattanayak^{3,4,5} · Marc Jeuland^{3,5,6,7}

Accepted: 18 September 2018 / Published online: 26 September 2018 © Springer Nature B.V. 2018

Abstract

Diarrheal illnesses and acute respiratory infections are among the top causes for premature death and disability across the developing world, and adoption of various technologies for avoiding these illnesses remains extremely low. We exploit data from a unique contingent valuation experiment to consider whether households in rural Rajasthan are unwilling to make investments in "domain-specific" environmental health technologies when faced with health risks in multiple domains. Results indicate that demand for water-related risk reductions is higher on average than demand for air-related risk reduction. In addition, households' private health benefits from mitigating diarrheal (respiratory) disease risks are higher (no different) when community-level air pollution risks, rather than community-level water pollution risks, have previously been mitigated. This asymmetric response cannot fully be explained by survey order effects or embedding, but rather suggests that that the broader health environment and the salience of particular risks may be important in households' decision to adopt environmental health technologies.

Keywords Household air pollution \cdot Diarrheal diseases \cdot Technology adoption \cdot Contingent valuation

JEL Classification Q51 · Q53 · Q56

1 Introduction

Diarrheal illnesses and acute respiratory infections continue to rank among the top causes for premature death and disability across South Asia (Institute for Health Metrics and Evaluation 2013). In India alone, it is estimated that diarrheal and respiratory illnesses are responsible

Marc Jeuland marc.jeuland@duke.edu

We thank the joint research program between Duke University and the Indian Institute of Management in Udaipur, which provided financial support for this study. Seva Mandir and Chitra provided support with field data collection. We also thank seminar participants at Duke University and the University of North Carolina Water and Health Conference for valuable comments. Usmani is grateful to the Duke Global Health Institute and the Duke University Energy Initiative for their generous support. Any remaining errors are our own.

Extended author information available on the last page of the article

for up to 6% of deaths (World Health Organization 2015). Children under five are particularly vulnerable to these infections, and there has long been speculation about the incidence of dual or sequential infections due to children's lesser immune function and the high incidence of malnutrition in at-risk communities (Guerrant et al. 2012; Walker et al. 2013).

Two important facts about environmental health conditions such as diarrhea and many respiratory illnesses are, first, that these conditions are preventable and, second, that many of the technologies that facilitate prevention are low-cost, readily available, and easy to use. For reducing the risk of diarrheal illnesses, point-of-use (POU) technologies such as boiling, liquid hypochlorite solution, disinfectant powders, solar disinfection, and various types of filtration devices are effective in combatting water contamination at the household level (Clasen et al. 2007; Fewtrell et al. 2005).¹ Meanwhile, household air pollution can be reduced through the use of improved cookstoves (ICS) or modern fuels (Anenberg et al. 2013). When compared with traditional biomass stoves, more efficient cooking technologies not only cut down on harmful emissions, they also decrease cooking times and use less fuel (Bensch et al. 2015; Bensch and Peters 2015; Brooks et al. 2016).

Despite these positive expected impacts, adoption and sustained use of environmental health technologies remains hard to achieve in poor, rural, and vulnerable communities (Jeuland et al. 2015d; Lewis and Pattanayak 2012; Whittington et al. 2012). Lack of demand among the rural poor is especially problematic as such households often face a greater burden from preventable illnesses (Braveman 2006).

At the same time, it is undeniable that poor households face myriad competing demands for limited resources (Banerjee and Duflo 2007). These competing demands may be especially salient in the health domain if households are unsure how to prioritize reductions in risks of different illnesses (Dow et al. 1999). Furthermore, the inability to save money or pay for the full up front cost of a technology may make households less likely to adopt preventative health products (Dupas and Robinson 2013). Complicating matters further, households typically consider non-health motives in deciding to adopt and use new technologies such as ICS or water treatment (Pattanayak and Pfaff 2009; Poulos et al. 2012; Thurber et al. 2013). Various factors such as convenience of use and changes in the taste of water or food can also make uptake of these and other challenges, it is not uncommon for the use of various health technologies (e.g., bed nets, ICS, or water treatment products) to diminish over time due to broken equipment, lack of replacement of consumables, or simple reversion back to former behaviors once promotion interventions have ended (Hensher et al. 2005; Luoto et al. 2011).

This existing literature clearly illustrates the myriad significant barriers to adoption of technologies for improving environmental health. Achieving greater prevention behavior thus presents important logistical challenges, and suggests a need for gaining a thorough understanding of consumer preferences for different solutions, prior to implementing efforts to stimulate demand. Needless to say, understanding what drives demand for preventive technologies is imperative to ensure future success in implementation, marketing, and behavior change or educational campaigns, and for improving health outcomes.

¹ POU water treatment technologies have gained prominence largely due to the continuing high reliance in many rural and low-income settings on community sources located outside the home. Water obtained from such sources varies in quality, and there is strong empirical evidence that water safety can be compromised between the time of collection and the time of consumption due to transport and storage, even when source water quality is fairly good. For example, Gasana et al. (2002) find that water tested at various source points indicated no risk to consumers, but that contamination levels were significantly higher at the point of use; there is similar evidence from other settings (Kremer et al. 2011; Rufener et al. 2010). In addition, investments in water infrastructure may also crowd out preexisting private prevention behavior, offsetting expected health benefits (Bennett 2012; Jessoe 2013; Jeuland et al. 2015b).

User preferences can be studied *ex ante* using a number of different approaches. Discrete choice experiments, for example, can be used to study how potential beneficiaries prioritize specific features of environmental health products (Jeuland et al. 2015a; Poulos et al. 2012). In this paper, however, we are not particularly interested in describing the demand for specific preventive technologies, which is already the subject of a substantial literature that includes, among other papers, several of those cited above. Instead, our objective is to better understand household beliefs with respect to competing risks.

Specifically, we study the relative perception of reduced health risks that would stem from hypothetical air and water quality improvements. To achieve this objective, we use the contingent valuation method (CVM), which is a common technique for determining the value of goods that are not bought and sold in ordinary markets (Hanemann 1994). This approach allows for estimation of a household demand curve for hypothetical improvements by randomizing prices across survey respondents and observing respondent willingness to pay at those different prices. While the CVM was initially introduced for valuing non-market environmental commodities, it is increasingly common in economic evaluations of healthcare and health improvement (Kartman et al. 1996; Lucas et al. 2007; Martín-Fernández et al. 2013). Previous work has stressed the importance of careful CVM study design and implementation, and critiques of the method are widespread (Carson 2000; Whittington 2002). Despite the well-known limitations of the CVM, it is appropriate for use in our study because it allows us to focus respondents' attention on specific changes in the risks of two types of illnessesrespiratory infections and diarrheal disease. Given that the typical implementation strategy (e.g., followed by government or NGOs) is to focus on interventions that target isolated problems, the CVM offers greater flexibility for studying the problem of complementarities in different types of health risk reductions.

Our study fits into a relatively broad literature that considers the willingness to pay (WTP) for environmental health improvements in low-income countries. Much of this prior literature is specific to measuring demand for household water treatment (Null et al. 2012; Orgill et al. 2013; Orgill-Meyer et al. 2018; Poulos et al. 2012; Van Houtven et al. 2017), with comparatively fewer studies—mostly discrete choice experiments—focusing on demand for technologies that reduce household air pollution (Jeuland et al. 2015a; Mobarak et al. 2012; van der Kroon et al. 2014). To the best of our knowledge, no previous study has explored WTP for these technologies in tandem, although public health researchers have considered the impacts of combined ICS and water treatment interventions (Rosa et al. 2014). Given that there is significant overlap in the burdens of disease due to diarrheal and respiratory illnesses in low-income contexts (as shown in Fig. 1), this is a prominent gap in our knowledge regarding the benefits of potentially lifesaving health products for the world's poor. In addition, our study design explores the extent to which the relative sequencing of the air and water quality solutions impacts stated WTP, which may shed light on the consistency of survey order effects or perceived complementarities across such hypothetical interventions.

This paper proceeds as follows. Section 2 presents a background on the drivers of demand for technologies that reduced water- and household-air-pollution-related risks, and discusses results from the existing literature on "order effects" in CVM studies, which are relevant for our work because we present respondents with offers for multiple hypothetical improvements in sequence. Section 3 outlines our study design and theoretical framework. Section 4 provides an overview of our sample, and presents key results. Section 5 presents the willingness to pay estimates obtained from these results. Section 6 features a discussion of our results, and Sect. 7 concludes.



Fig. 1 Global burden of disease due to diarrhea and respiratory infections **a** Diarrheal diseases, Disability Adjusted Life Years (DALYs) per 100,000 and **b** Respiratory infections, Disability Adjusted Life Years (DALYs) per 100,000. Burden of disease data obtained from the World Health Organization (2014)

2 Background

Many researchers have examined the set of socioeconomic and demographic drivers of demand for technologies that reduce water- and air-related disease risks. Lewis and Pattanayak (2012) synthesize the existing empirical evidence on determinants of ICS and clean-fuel adoption and find a "systematic and theoretically consistent relationship between adoption of clean energy products and socioeconomic status (including income, education, and social marginalization) and urban location." These findings are largely consistent with the correlates identified by Rehfuess et al. (2013), who also include qualitative analyses and case studies in their review. Others have studied the important role played by household preferences (Ouedraogo 2006); social networks (Miller and Mobarak 2015); use-based incentives (Usmani et al. 2017); and the relative availability and affordability of substitute technologies or fuels (Beyene and Koch 2013).

Demand studies that focus on POU water treatment technologies highlight a similar set of factors. Despite their low cost and efficacy, poverty and lack of education pose significant barriers to adoption (Freeman et al. 2009; Hunter 2009). Subsidies can be effective at increasing adoption (Null et al. 2012), though the role of positive prices in screening out those most likely to actually use POU technology remains a matter of some debate (Ashraf et al. 2010). Experimental evidence suggests that a lack of knowledge about contamination risks may also suppress demand, particularly among households with low socio-economic status (Brown et al. 2016; Hamoudi et al. 2012; Orgill et al. 2013). In addition, there are signs that additional unrecognized barriers exist beyond "traditional" constraints related to income and information (Luoto et al. 2011). In particular, preferences for POU technologies have been found to be heterogeneous, and distaste or inconvenience often appears to discourage adoption (Jeuland et al. 2015c). Research also points to the importance of psycho-social (behavior change) interventions that activate households' awareness of the benefits of POU technologies in new and effective ways (Blanton et al. 2010; Boisson et al. 2010).

As mentioned above, although the prior literature has analyzed demand for each of household water treatment and household air pollution reduction technologies separately, we are not aware of any study that has looked at WTP for the two sets of technologies in tandem. An important challenge in conducting such a study of two technologies-which we address by randomizing the order in which the two technologies are presented to each respondent-arises from the fact that stated preference estimates have been found to be sensitive to order effects, whereby the sequencing of valuation questions affects the propensity to respond affirmatively to a given price offer (Halvorsen 1996). For instance, Samples and Hollyer (1990) find that survey respondents' willingness to pay to preserve seals from extinction is greatly affected by whether it is elicited before or after that for whales. Similarly, Powe and Bateman (2003) investigate the willingness of visitors to pay to prevent saline flooding in a wetland area in eastern England; they find that estimated WTP for a smaller component of the program is significantly higher if it is elicited before that for the larger component. While CVM surveys that elicit WTP for multiple projects or policies appear to yield lower estimates for items further down in the sequence (Hoehn and Loomis 1993), this is not always so (Clark and Friesen 2008). A variety of explanations have been advanced for the presence of CVM order effects, including strategic misrepresentation of WTP by respondents (Clark and Friesen 2008); respondents' desire to purchase "moral satisfaction" (Kahneman and Knetsch 1992); and the extent to which the goods being valued are substitutes (Carson et al. 1993). More recent work has also found that adjusting the way stated preference surveys are executed can reduce WTP estimates (Cook et al. 2011; Longo et al. 2015).

3 Methodology

3.1 Sample Selection and Study Design

Our data come from a baseline household survey conducted in collaboration with a local non-governmental organization between August and October 2013 in the Udaipur district of Rajasthan, India. The final sample included 900 households living in sixty villages; fifteen households were randomly selected to participate in each village, using a field-based counting method. In our analysis, results from 38 households were dropped due to incomplete surveys.

The household survey instrument included questions on household composition, demographics, and socioeconomic status; respiratory illness and diarrheal disease prevalence; fuel and water sources; water treatment, storage, and hygiene behaviors; WTP for reduced risk of respiratory or diarrheal illness; as well as detailed perceptions of environmental risks including those associated with poor air and drinking water quality. Table 1 presents a description of some of the key variables collected through the survey, while Table 2 presents descriptive statistics for the 862 households included in our final analytical sample.

As noted before, the demand for preventative health products incorporates dimensions such as convenience of use, reliability, aesthetics, and cost. The purpose of the CVM exercise Table 1 Descriptions of key variables

VARIABLE	DESCRIPTION
ACCEPTWATER	= 1 if accepted offer for water purifier
ACCEPTAIR	= 1 if accepted offer for air purifier
PRICE	Price offer (current US dollars)
WATERFIRST	= 1 if WTP offer option "A" (community ARI scenario + option to buy water filter first)
PRICE×WATERFIRST	$PRICE \times WATERFIRST$
HHSIZE	Household size
CHILD5	Number of children under five
FEMALEHEAD	= 1 if female-headed household
CASTE	= 1 if classified as SC/ST/NT/OBC ("lower caste")
HINDU	= 1 if Hindu
EDUCATION	Average minimum years of education (adults in household)
EDUCATIONFEMALE	Average minimum years of education (females in household)
EDUCATIONMALE	Average minimum years of education (males in household)
ROOMS	Number of rooms
FARMER	= 1 if household head employed in agriculture
POVLINE	= 1 if below poverty line
ELECTRICITY	= 1if access to electricity "all the time" or "sometimes"
HEARDICS	= 1 if heard of improved cookstoves (ICS)
NEGSTOVE	= 1 if heard of negative impacts of cooking smoke
EXPENDITURE	Monthly expenditure (current US dollars)
VILLCLEAN	= 1 if cleanliness of village "very clean" or "clean"
ARIFATAL	= 1 if respiratory illness indicated as "most dangerous" (fatal)
REGULARICS	= 1 if ICS used at least "3–4 times per week"
CLEANFUEL	= 1 if clean fuel used (not necessarily exclusively)
IMPROVEDWATER	= 1 if main drinking water source is "improved" (WHO definition)
UNSAFEWATER	= 1 if unsafe method of pouring drinking water demonstrated
HEARDWASH	= 1 if received WASH public health message in the past
WATERSAFETY	Drinking water safety perception (10 = completely safe; 0 = completely unsafe)
SMOKESAFETY	Stove smoke safety perception (10 = completely safe; 0 = completely unsafe)
DIARRPREVENT	= 1 if believes diarrhea can be prevented
COOKTIME	Total time spent cooking per day (hours)
COLLTIME	Total time spent collecting non-clean fuels per week (hours)
WATERCOLL	Total time spent collecting water per week (hours)
TOTALCOUGH	Number of household members who have suffered from cough/cold in the past two weeks
COUGHPAY	Total amount spent to treat most recent bout of all members' cough/cold (current US dollars)
TOTALDIARR	Total household days lost to most recent diarrheal episode (caring for someone or sick with diarrhea)

VARIABLE	DESCRIPTION
DIARRPAY	Total amount spent to treat most recent bout of all members' diarrheal illness (current US dollars)
CHILDDIARR	= 1 if any incidence of childhood diarrhea in household in the past two weeks
TOTALMALARIA	= 1 if any incidence of malaria in the past one year
ННТВ	= 1 if any incidence of TB in the past

Table 1 continued

Table 2 Descriptive statistics

	MEAN	MEDIAN	SD	MIN	MAX
ACCEPTWATER	0.51	1	0.50	0	1
ACCEPTAIR	0.48	0	0.50	0	1
OPTION	0.49	0	0.50	0	1
PRICE	18.9	16.3	14.7	4.88	40.7
HHSIZE	5.63	6	2.00	1	15
CHILD5	0.43	0	0.66	0	4
FEMALEHEAD	0.033	0	0.18	0	1
CASTE	0.93	1	0.25	0	1
HINDU	1.00	1	0.034	0	1
EDUCATION	2.92	3	2.73	0	12
EDUCATIONMALE	4.09	3.50	3.76	0	21
EDUCATIONFEMALE	1.56	0	2.72	0	12
ROOMS	5.07	5	1.34	1	10
FARMER	0.53	1	0.50	0	1
POVLINE	0.78	1	0.42	0	1
ELECTRICITY	0.63	1	0.48	0	1
HEARDICS	0.59	1	0.49	0	1
NEGSTOVE	0.98	1	0.15	0	1
EXPENDITURE	99.1	87.8	59.6	17.9	887.8
VILLCLEAN	0.46	0	0.50	0	1
ARIFATAL	0.064	0	0.25	0	1
REGULARICS	0.068	0	0.25	0	1
CLEANFUEL	0.84	1	0.37	0	1
IMPROVEDWATER	0.56	1	0.50	0	1
UNSAFEWATER	0.71	1	0.46	0	1
HEARDWASH	0.99	1	0.084	0	1
WATERSAFETY	6.45	6	1.94	0	10
SMOKESAFETY	4.57	5	1.88	0	10
DIARRPREVENT	0.98	1	0.15	0	1
COOKTIME	3.25	3	1.79	0	33
COLLTIME	15.0	10	13.4	0	105.5
WATERCOLL	5.28	4.67	3.99	0	25.1

	MEAN	MEDIAN	SD	MIN	MAX
TOTALCOUGH	1.52	1	1.40	0	12
COUGHPAY	8.74	3.25	22.8	0	487.8
TOTALDIARR	2.40	0	5.42	0	53
DIARRPAY	4.44	0	15.0	0	308.9
CHILDDIARR	0.065	0	0.25	0	1
TOTALMALARIA	0.72	1	0.45	0	1
ННТВ	0.022	0	0.15	0	1
Observations	855				

Table 2 continued

carried out in the survey was to measure WTP for diarrheal risk and respiratory risk reductions that would not be contaminated by these other features. As such, respondents were asked to imagine a hypothetical water purification technology and a hypothetical air purification technology that was costly but that would not affect other features of their drinking water or air supply.

3.2 The Contingent Valuation Experiment

Prior to survey administration, households were randomly assigned to two different option groups for the CVM scenario (i.e., 450 households each received an "Air framing" [Option A] or a "Water framing" [Option B]). For those assigned to Option A, the CVM scenario first asked households to imagine that an air quality program had been implemented in their community that would cost them a nominal amount of INR 50 (approximately \$0.80 at the time) per year; knowing this information, they were then asked if they would also pay a randomly assigned annual price (ranging from INR 300 to INR 2500) to acquire and use a private household-level water purification device that would reduce the risk of diarrhea. For households assigned to Option B, the scenario was inverted: respondents were first asked to imagine a community-level water improvement program that had cost them INR 50 per year, and were then asked if they would also pay a randomly assigned price for a private household-level air purification device that would reduce the risk of respiratory illness.

In both the air (Option A) and water (Option B) frames, following responses to this first WTP question for the private improvement that would reduce the other disease, respondents were then asked a second WTP question about the private device that would reduce the risk of the other illness (a household-level air purifier in Option A, and a water purifier in Option B), which would further reduce risks that had already been reduced by the community-level intervention in that frame. In this second WTP question, the same randomly assigned price given for the first private device that targeted the other disease was again used for each household. The motivation for including this second question was to test for order effects or changes in the marginal value of additional risk reductions. Each of these could lead to systematic differences in the WTP for a purification device depending on whether it was offered first or second, and could shed light on perceived complementarities in disease risk reductions. Figure 2 provides a visual overview of our CVM experiment scenarios and ordering. The full scenario script and sample visuals that were used to illustrate the risk reduction concepts to respondents are included in Appendix A.

1081







Fig. 2 Overview of contingent valuation experiment scenarios and ordering. Survey respondents were randomly assigned to one of the two outlined options

3.3 Theoretical Framework and Model Specification

Estimating separate univariate probit regressions for the decisions to accept the air and water purifier offers would ignore any overlap in unobserved characteristics that impact both purchase decisions (Greene 2011). We, thus, employ a bivariate probit regression approach to more consistently consider the association between household-level characteristics and demand. Explanatory variables for WTP were chosen based on findings from previous work that speaks to the demand for preventive health improvements (Lucas et al. 2007). For instance, households with young children—who are especially susceptible to diarrheal and respiratory illnesses—or those that believe these diseases can be prevented may be more willing to pay for health improvements. On the other hand, households that perceive their current environmental quality to be safe, or that do not believe that these illnesses can be prevented, may be less willing to pay. Thus, the determinants included in our model encompass a variety of socioeconomic and demographic characteristic, as well as any current knowledge or behaviors for reducing disease risks.

Our empirical model is specified as follows:

$$Y_{i,j} = \beta_{0,j} + \beta_{1,j} PRICE_i + \beta_{2,j} WATERFIRST_i + \beta_{3,j} (PRICE_i \times WATERFIRST_i) + \sum_n \beta_{n,j} X_{n,i} + \epsilon_{i,j}$$
(1)

Equation (1) represents a bivariate probit model that explores the drivers of household *i*'s decisions regarding purchasing technology *j* (an air or water purifier). $Y_{i,j}$ is a binary variable that equals one if household *i* accepted the offer for technology *j* and zero otherwise; $PRICE_i$ is the randomized price offer faced by household i for both technologies;

WATERFIRST_i is a binary variable that is equal to one if household *i* is randomly assigned Option A (the air frame, in which the first private technology was a household water purifier) and zero if the assignment is Option B; $X_{n,i}$ represents a series of n household-level socioeconomic, demographic, and behavioral controls (that is informed by the prior literature on the correlates of demand); and $\beta_{0,j}$ is a constant. We estimate this model for technologies *j* and *j'*; in line with our bivariate probit specification, the unobserved terms $\epsilon_{i,j}$ and $\epsilon_{i,j'}$ are jointly distributed as standard bivariate normal with correlation ρ .

In such a specification, the coefficients indicate the average effect of each of the explanatory variables on the probability of accepting the relevant price offer. Note, however, that we also interact the randomized price offer with the dichotomous variable indicating the household's option assignment.

The results of this estimation allow us to consider the relative importance of disease risk reduction complementarities relative to that of more commonly explored order and "embedding" effects, where the latter refers to the fact that stated WTP for a good may vary depending on whether it is evaluated on its own or as part of a more inclusive category (Kahneman and Knetsch 1992). In our case, this effect could arise from the fact that individuals were told to consider that they had already contributed to a health-risk reduction in one of the health domains through a mandatory community program, which might have led them to have systematically different willingness to pay for an additional health risk reduction in that domain, relative to the other second domain.

Thus, when $WATERFIRST_i = 0$ (the water frame, where the first private technology that is offered is an air purifier), our model reduces to

$$Y_{i,j} = \beta_{0,j} + \beta_{1,j} PRICE_i + \sum_n \beta_{n,j} X_{n,i} + \epsilon_{i,j}.$$
(2)

If, instead, $WATERFIRST_i = 1$ (the air frame, where the first private technology that is offered is a water purifier), the empirical model becomes

$$Y_{i,j} = (\beta_{0,j} + \beta_{2,j}) + (\beta_{1,j} + \beta_{3,j}) PRICE_i + \sum_n \beta_{n,j} X_{n,i} + \epsilon_{i,j}.$$
 (3)

The coefficient $\beta_{2,j}$ may thus be interpreted as the marginal change in the probability of adopting technology *j* at a price of zero under the air frame (Option A). Similarly, the coefficient $\beta_{3,j}$ represents the marginal change in responsiveness of probability of accepting an offer to the product's price in the same situation. Several CVM studies—particularly those employing a dichotomous choice elicitation approach, where respondents are asked to simply accept or reject a particular proposal, as we do—have uncovered evidence of "order effects," where offering respondents a particular option first yields relatively higher WTP estimates (Boyle et al. 1993; Clark and Friesen 2008). However, as noted before, we are not aware of any attempt to simultaneously frame the elicitation question within the context of a related disease risk. For this reason, our expectations regarding the signs of $\beta_{2,j}$ and $\beta_{3,j}$ are not clear-cut.

4 Results

4.1 Sample Description

Table 2 presents descriptive statistics for our study sample of 862 households. The average household size is 5.6 people, and households are generally poor: nearly 80% report being

below the poverty line, with an average monthly household expenditure across the sample of approximately \$100. In addition, over 90% of households identify as belonging to one of India's historically disadvantaged castes or tribes. Only about 3% are headed by females. Access to electricity in the area appears to be low and unreliable, with only about 60% of respondents stating that they receive electricity at least "sometimes." Unsurprisingly, in the absence of dependable access to modern energy services, households spend on average fifteen hours per week collecting solid or other non-clean fuels every week. Combined with the time dedicated to collecting water, this burden is in excess of twenty hours per week.

Sample households also appear to be navigating a burdensome disease environment. For instance, nearly 7% of respondents report at least one case of childhood diarrhea in the two weeks prior to survey, while one in four household members report suffering from cough or cold symptoms over the same period. Households reporting illnesses spent approximately 11 and 9% of monthly expenditure to treat members' most recent episodes of cough/cold and diarrhea, respectively, and spent nearly five days sick with—or caring for someone suffering from—diarrhea.² Despite this nontrivial burden, rates of health-risk-averting behavior are low. For instance, only about 7% of respondents stated that they used an improved cookstove at least three to four times per week. Similarly, 70% of respondents did not take adequate measures to prevent their hands from coming into contact with stored drinking water in a household demonstration, as recommended by international drinking-water-quality guide-lines (World Health Organization 2011).

The dearth of health-risk-averting behavior does not appear to be accompanied by a lack of knowledge in this sample. For example, 60% of households claim to have heard of ICS, while nearly all indicate that cooking smoke has negative effects. Similarly, almost all households state that diarrhea is preventable, and report having received a public health message related to water, sanitation, or hygiene in the past. Perceptions related to one's own drinking water and household air pollution levels, however, appear to be mixed: on an integer scale of one to ten—with ten representing "completely safe"—households identify the safety of their drinking water and household stove smoke as 6.4 and 4.6, respectively. The salience of individual diseases in respondents' minds also appears to be skewed: a surprisingly low proportion of respondents (just 6%) identified acute respiratory infections as more likely to be fatal, while nearly three times as many said the same for diarrhea.³ These differences in perceptions related to health risks are likely to impact perceived benefits from making related risk-reduction investment and, thus, are a key motivation for this paper.

4.2 Estimates for Respiratory and Diarrheal Disease Risk Reductions with Simple Order Effects

We first run our WTP regressions allowing only for *constant shifts* in demand under different framing (i.e., excluding the $PRICE \times WATERFIRST$ interaction term that allows for differential price sensitivity). Columns 1–3 of Table 3 report results for the private air purifier, and columns 4–6 report the results for the private water purifier; all reported results explicitly allow for correlation in the water and air purifier purchase decisions. In model (1) of Table 3,

 $^{^2}$ About 20% of households reported no instances of cough or cold in the two weeks prior to the survey, while around half reported no instances of diarrhea over the same period. Averaging over all households, approximately 9 and 4% of monthly expenditure were spent treating the most recent bouts of cough/cold and diarrhea, respectively.

³ The remainder (approximately 75%) stated that acute respiratory infections and diarrheal diseases are "equally dangerous."

we include only *PRICE* and *WATERFIRST* as explanatory variables. Consistent with economic theory, we find a negative and highly statistically significant relationship between *PRICE* and the probability of accepting the air purifier offer. There is some evidence for an order effect, but it is not consistent across technologies. Respondents asked about the purchase of a water purifier before the air purifier are more likely to respond affirmatively to the offer for the former than those asked about the air purifier first. Meanwhile, those asked about the water purifier first are somewhat less likely to respond affirmatively to the offer for the air purifier, but that effect is weaker and not distinguishable from zero.

In models (2) and (3), we sequentially introduce an increasing number of controls related to household-level socioeconomic and demographic characteristics (in model 2); and then add disease and risk perceptions; current behavior related to safe or unsafe stove and water use; time spent cooking, or collecting fuel and water; and household history regarding other diseases (in model 3). The sign and significance of the price coefficient remains stable, as would be expected given that this was randomized at the household level. In addition, the inclusion of these controls leads to the coefficient for WATERFIRST becoming significant at the 10% level in columns (2) and (3), hinting at the presence of a weak order effect. A number of other consistent patterns emerge.⁴ We find that households with a higher average minimum level of adult education are more likely to accept the air purifier purchase offer. The more basic specification in model (2) reveals a positive relationship between the air purifier purchase decision and the number of rooms (a proxy for wealth). Finally, some practices and perceptions appear to play a role; households stating that they use clean fuels (CLEANFUEL) are more likely to purchase air purifiers, despite arguably "needing" them less, as are those that believed that respiratory illnesses are "most dangerous" (ARIFATAL) compared to diarrheal diseases.

Columns 4–6 reports the results from an identical set of regressions for the water purifier, once again controlling for constant, price-invariant, order effects. A similar story emerges with one notable exception, in that we observe a clear, consistent, and significant order effect in this case. Specifically, we find that *WATERFIRST* is a positive and statistically significant predictor of the probability of accepting the offer. That is, households are more likely to accept the water purifier offer if they are presented with this technology before they receive the offer to purchase a private air purifier. Otherwise, higher prices make it less likely that households accept the water purifier offer; and more educated and larger households are more likely to accept the offer, as are households with a larger number of rooms. Practices and perceptions continue to matter—*CLEANFUEL* and *ARIFATAL* are, as before, positively related to the probability of accepting the offer.⁵

These results suggest that there is some evidence of a positive order effects (i.e., higher likelihood of saying yes to a price offer for initial offers than for subsequent improvements), but that the pattern is inconsistent, and appears to vary with the type of risk reduction. We return to this issue in Sect. 6.

⁴ These patterns are entirely consistent when different groups of variables are excluded from model (3) (results not shown).

⁵ Interestingly, neither of the water-related perceptions variables (IMPROVEDWATER and UNSAFEWATER) are statistically significant predictors of the water purifier purchase decision, although they do have the "expected" signs. These results for perception variables suggest that clean-fuel use and consideration that respiratory diseases are particularly dangerous may simply be correlated with WTP for environmental health improvements in general, rather than indicating something specific about differential demand for respiratory versus diarrheal disease risk reductions.

Table 3 Bivariate probi	t regression results for V	VTP for air and water puri	ifiers, with price-invaria	ant order effects		
VARIABLES	(1) ACCEPTAIR	(2) ACCEPTAIR	(3) ACCEPTAIR	(4) ACCEPTWATER	(5) ACCEPTWATER	(6) ACCEPTWATER
PRICE	-0.0226^{***}	-0.0232^{***}	-0.0233^{***}	-0.0253^{***}	-0.0267^{***}	-0.0272^{***}
	(0.00391)	(0.00390)	(0.00394)	(0.00318)	(0.00314)	(0.00320)
WATERFIRST	-0.169	-0.178^{*}	-0.182^{*}	0.211^{**}	0.212^{**}	0.231^{**}
	(0.106)	(0.103)	(0.105)	(0.0937)	(0.0908)	(0.0903)
HHSIZE		0.0175	0.0213		0.0559^{**}	0.0628^{***}
		(0.0239)	(0.0259)		(0.0223)	(0.0234)
FEMALEHEAD		0.0339	0.155		-0.297	-0.225
		(0.241)	(0.241)		(0.258)	(0.280)
EDUCATION		0.0733^{***}	0.0734^{***}		0.102^{***}	0.0964^{***}
		(0.0178)	(0.0189)		(0.0143)	(0.0155)
ROOMS		0.109^{***}	0.0718^{*}		0.0786^{*}	0.0612
		(0.0398)	(0.0405)		(0.0418)	(0.0428)
FARMER		0.0268	0.0852		0.0153	0.0494
		(0.104)	(0.105)		(0.0916)	(0.0941)
POVLINE		0.121	0.108		0.0171	0.00719
		(0.106)	(0.113)		(0.112)	(0.117)
ELECTRICITY		0.102	0.0166		0.0599	-0.00546
		(0.104)	(0.117)		(0.117)	(0.119)
HEARDWASH			-0.826			-0.159
			(0.629)			(0.519)
SMOKESAFETY			-0.00837			-0.0356
			(0.0251)			(0.0244)
VILLCLEAN			0.122			0.119
			(0.103)			(0.0981)

Table 3 continued						
VARIABLES	(1) ACCEPTAIR	(2) ACCEPTAIR	(3) ACCEPTAIR	(4) ACCEPTWATER	(5) ACCEPTWATER	(6) ACCEPTWATER
ARIFATAL			0.368^{**}			0.447***
			(0.187)			(0.164)
REGULARICS			0.245			0.307
			(0.226)			(0.244)
CLEANFUEL			0.268^{*}			0.325^{**}
			(0.142)			(0.141)
IMPROVEDWATER			0.144			0.0266
			(0.104)			(0.110)
UNSAFEWATER			-0.276^{***}			-0.0537
			(0.102)			(0.117)
COOKTIME			-0.0153			-0.0111
			(0.0269)			(0.0180)
COLLTIME			0.00710^{*}			0.00402
			(0.00380)			(0.00385)
WATERCOLL			-0.00405			0.0154
			(0.0120)			(0.0120)
HHTB			0.0361			-0.257
			(0.292)			(0.292)
CHILDDIARR			0.197			-0.0119
			(0.187)			(0.172)
Constant	0.449^{***}	-0.581^{**}	0.221	0.394^{***}	-0.640^{**}	-0.678
	(0.102)	(0.256)	(0.668)	(0.0893)	(0.307)	(0.619)
Observations	855	855	855	855	855	855
Standard errors—in parenth	eses-clustered at the	village level. *** $p < 0.0$	01, ** p < 0.05, * p <	0.1		

 Δ Springer

1087

4.3 Adding Price-Varying Order Effects

In Table 4, we report results for regressions similar to those reported in Table 3, except that we additionally allow for the order effects to vary with price (i.e., we include a $PRICE \times WATERFIRST$ interaction). The inclusion of this interaction term yields additional insights. Note first the difference in magnitudes of the coefficient on PRICE between the models shown in Table 3 and those in Table 4. As the inclusion of the $PRICE \times WATERFIRST$ interaction term allows us to identify changes in the relationship between the price and the probability of accepting the offer based on which device was offered first, the interpretation of the estimated coefficients changes, as shown in equations (2) and (3). Namely, in the case of the air purifier (water purifier), the coefficient $\beta_{PRICE}^{AIR} = -0.0188$ $(\beta_{PRICE}^{WATER} = -0.0208)$ is the price responsiveness of the probability of accepting the relevant offer when the private air purifier is offered first. This price sensitivity is considerably lower than that observed when the private water purifier is offered first; in that case the price sensitivity for the air (water) purifier is $\beta_{PRICE}^{AIR} + \beta_{PRICE\times WATERFIRST}^{AIR} = -0.0283$ $(\beta_{PRICE}^{WATER} + \beta_{PRICE \times WATERFIRST}^{WATER} = -0.0342)$. That is, the probability of accepting the offer for both devices is more price responsive when an offer to purchase the water filter comes first.

As noted before, the interpretation of the estimated coefficient for WATERFIRST also changes with the inclusion of the interaction term; it is the marginal change in the probability of accepting the price offer if the price is zero. We find that $\beta_{WATERFIRST}^{WATER}$ is large, positive, and statistically significant. At the same time, while the coefficient on $\beta_{WATERFIRST}^{AIR}$ is negative, its magnitude is considerably smaller and it is never statistically significant. Together, this suggests that the air framing (i.e., when respiratory risks have been reduced by a community intervention) makes households *more* likely to purchase the water purification technologies without having a statistically meaningful impact on the corresponding demand for air purification technologies.

Collectively, these results provide evidence of a somewhat more nuanced order effect than that first uncovered in Table 3. Namely, it suggests a positive shift in the probability of accepting a price offer for technologies that reduce health risks broadly when the private water purifier is offered first. Specifically, in the case of the air this positive shift is almost exactly offset by a higher price sensitivity and the net effect on demand is weak at best. In contrast, for the water purifier, the positive shift is larger and the change in the price sensitivity is lower, such that a larger net increase in demand is apparent.

Most of the other results previously observed are robust to the inclusion of the interaction term. *CLEANFUEL* and *ARIFATAL* remain statistically significant and positive, with magnitudes that are largely the same; the same holds true for *EDUCATION*. The robustness of these findings across specifications increases our confidence in the stability of these correlations, and in the randomization of prices and question order across respondents.

5 Deriving Willingness to Pay

We can use our estimates to derive households' WTP for the proposed air purifier and water purifier technologies. Following Haab and McConnell (2002), given our basic probit specification (without a price-order interaction term), mean willingness to pay for each technology j (air purifier or water purifier) may be expressed as

$$WTP_{j} = -\frac{\beta_{0,j} + \beta_{2,j} + \sum_{n} \beta_{n,j} X_{n}}{\beta_{1,j}}.$$
(4)

🖉 Springer

Table 4 Bivariate probit regress	sion results for WTP for	or air and water purifier	s, allowing for price-v	arying order effects		
VARIABLES	(1) ACCEPTAIR	(2) ACCEPTAIR	(3) ACCEPTAIR	(4) ACCEPTWATER	(5) ACCEPTWATER	(6) ACCEPTWATER
PRICE	-0.0189^{***}	-0.0187^{***}	-0.0188^{***}	-0.0196^{***}	-0.0203^{***}	-0.0208***
	(0.00451)	(0.00444)	(0.00439)	(0.00463)	(0.00459)	(0.00462)
WATERFIRST	-0.0249	-0.00420	-0.00867	0.432^{***}	0.457^{***}	0.479^{***}
	(0.163)	(0.158)	(0.167)	(0.166)	(0.159)	(0.162)
PRICE × WATERFIRST	-0.00784	-0.00946	-0.00946	-0.0119^{*}	-0.0132^{**}	-0.0134^{**}
	(0.00675)	(0.00665)	(0.00663)	(0.00646)	(0.00632)	(0.00643)
HHSIZE		0.0160	0.0189		0.0543^{**}	0.0599^{**}
		(0.0236)	(0.0258)		(0.0228)	(0.0238)
FEMALEHEAD		0.0558	0.180		-0.268	-0.191
		(0.241)	(0.241)		(0.263)	(0.286)
EDUCATION		0.0738^{***}	0.0743^{***}		0.102^{***}	0.0978^{***}
		(0.0175)	(0.0187)		(0.0143)	(0.0157)
ROOMS		0.112^{***}	0.0750^{*}		0.0822^{**}	0.0656
		(0.0394)	(0.0401)		(0.0415)	(0.0425)
FARMER		0.0339	0.0921		0.0252	0.0593
		(0.103)	(0.105)		(9060)	(0.0937)
POVLINE		0.120	0.106		0.0153	0.00402
		(0.106)	(0.113)		(0.112)	(0.117)
ELECTRICITY		0.105	0.0201		0.0654	0.00154
		(0.104)	(0.117)		(0.117)	(0.119)
HEARDWASH			-0.856			-0.194
			(0.638)			(0.552)
SMOKESAFETY			-0.00968			-0.0374
			(0.0250)			(0.0239)
VILLCLEAN			0.116			0.110
			(0.103)			(0.0984)

Table 4 continued						
VARIABLES	(1) ACCEPTAIR	(2) ACCEPTAIR	(3) ACCEPTAIR	(4) ACCEPTWATER	(5) ACCEPTWATER	(6) ACCEPTWATER
ARIFATAL			0.371^{**}			0.449^{***}
			(0.186)			(0.163)
REGULARICS			0.230			0.279
			(0.229)			(0.243)
CLEANFUEL			0.272^{*}			0.332^{**}
			(0.143)			(0.141)
IMPROVEDWATER			0.140			0.0212
			(0.104)			(0.110)
UNSAFEWATER			-0.282^{***}			-0.0616
			(0.101)			(0.116)
COOKTIME			-0.0128			-0.00778
			(0.0281)			(0.0173)
COLLTIME			0.00700^{*}			0.00390
			(0.00380)			(0.00379)
WATERCOLL			-0.00357			0.0160
			(0.0121)			(0.0120)
HHTB			0.0228			-0.281
			(0.288)			(0.289)
CHILDDIARR			0.195			-0.0137
			(0.187)			(0.173)
Constant	0.382^{***}	-0.675^{***}	0.163	0.292^{***}	-0.773^{**}	-0.772
	(0.116)	(0.262)	(0.682)	(0.110)	(0.311)	(0.644)
Observations	855	855	855	855	855	855
Standard errors—in parenth	eses-clustered at the	village level. *** $p < 0.0$	1, ** p < 0.05, * p <	0.1		

ingness to pay for purifier, USD per		$MEAN^{\dagger}$
er month	Air purifier, WATERFIRST unspecified [‡]	1.34
		(-0.33, 4.80)
	Water purifier, WATERFIRST unspecified [‡]	1.63
		(0.06, 4.97)
	Air purifier, $WATERFIRST = 1$	1.09
		(-0.34, 4.12)
	Air purifier, $WATERFIRST = 0$	1.68
		(-0.28, 6.38)
	Water purifier, $WATERFIRST = 1$	1.91
		(0.77, 4.50)
	Water purifier, $WATERFIRST = 0$	1.22
		(-0.77, 5.72)

[†]Bootstrapped 90% confidence intervals in parentheses. [‡]PRICE \times WATERFIRST interaction term not included in regression model

The inclusion of the interaction term somewhat modifies this expression. Specifically, when WATERFIRST = 1, mean willingness to pay is

$$WTP_{j} = -\frac{\beta_{0,j} + \beta_{2,j} + \sum_{n} \beta_{n,j} \bar{X}_{n}}{\beta_{1,j} + \beta_{3,j}}.$$
(5)

When WATERFIRST = 0, this expression reduces to

$$WTP_j = -\frac{\beta_{0,j} + \sum_n \beta_{n,j} X_n}{\beta_{1,j}}.$$
(6)

Note that implicit in these expressions is the assumption that the demand curve for technology *i* is exponential.⁶

Table 5 reports our estimates for mean WTP. The estimates reported in the first two rows are obtained using models (3) and (6) in Table 3, i.e., they do not include a $PRICE \times WATERFIRST$ interaction term in the underlying regression specification. These estimates yield a WTP of \$1.34 per household per month for the air purifier, an amount that is nearly 20% less than the willingness to pay for the water purifier (\$1.63).

While useful in terms of gauging households' average valuation of these technologiesand, consequently, their valuation for the related disease risk reductions-not incorporating the interaction term masks the considerable variations in WTP that arise from the order in which the improvements were offered to respondents. In rows 3-4 and 5-6 of Table 5, we calculate WTP estimates from models (3) and (6) from Table 4, respectively. Two key conclusions may be drawn. First, the pattern observed in the first two rows of Table 5 continues to hold, i.e., when compared across analogous ordering scenarios, it is observed that households are willing to pay a substantial premium for the water purifier over the air

Table 5 Will air and water household pe

⁶ We assume an exponential demand curve primarily for empirical tractability and, in particular, to enable comparability of our estimates with the related literature that looks at WTP for environmental quality in lowincome settings (e.g., Orgill et al. 2013). The results we present are qualitatively unchanged if we assume that demand for purification technologies is linear.

	Air purifier	Water purifier	Difference	(p-value)
WATERFIRST unspecified [‡]	1.34	1.63	-0.30	0.00
WATERFIRST = 1	1.09	1.91	-0.82	0.00
WATERFIRST = 0	1.68	1.22	0.46	0.00
WATERFIRST = 1/WATERFIRST = 0	1.09	1.22	-0.13	0.00
WATERFIRST = 0/WATERFIRST = 1	1.68	1.91	-0.23	0.00

Table 6 Paired t test results for mean willingness to pay estimates

[‡]PRICE \times WATERFIRST interaction term not included in regression model

purifier—specifically, up to 15% more.⁷ Households' WTP for the water purifier is higher than that for the air purifier.

Second, households are most willing to pay for *both* goods jointly when WATERFIRST = 1, though the difference is not large. In other words, in the context of a hypothetical respiratory-illness-related community-level intervention and an offer to purchase the water purifier first (Option A), household perceptions of private benefits from both the water and air purifier appear to be jointly maximized.

Table 6 presents results from paired *t* tests, which indicate that the difference between each of these mean WTP amounts is statistically significant. Table 6 also illustrates that the higher WTP for the water purifier in our basic specification (a statistically significant difference of \$0.30) appears to be largely driven by the higher WTP for the technology when WATERFIRST = 1 (\$0.82), rather than when WATERFIRST = 0 (\$0.46). As a further test of the robustness of our results, we generate 1000 bootstrapped estimates of each of our WTP measures. This allows us to examine the extent to which our results are driven by our specific study sample.⁸ The distributions of these bootstrapped estimates are shown in Fig. 3. A visual inspection of these distributions suggests that our results remain unchanged: mean WTP for the water purifier is higher than that for the air purifier (Fig. 3, panel a), and this higher WTP is almost entirely driven by the results in the air framing of Option A, which established a community-level respiratory risk reduction prior to the purchase offer for each of the private improvements (Fig. 3, panels b and c).

Note that while the nominal amounts may seem small, these are non-trivial sums in lowincome contexts. Individually, for instance, WTP for each technology is approximately 1-2%of monthly household expenditure; taken together as an air-and-water-purifier bundle, this amounts to approximately 2-4% of monthly household expenditure. As a comparison, a frequently-quoted measure of affordability in the case of piped water access in developing countries is the so-called "five-percent rule," that is, improvements in access to water services via the provision of piped water are considered affordable and appropriate if the cost to households is approximately 5% of income (McPhail 1993).

⁷ As the dichotomous variable *WATERFIRST* captures the assignment of a scenario that is "mirrored" across technologies, the appropriate comparisons to make across the WTP estimates for air and water purifier are as follows: Air Purifier, *WATERFIRST* = 1 with Water Purifier, *WATERFIRST* = 0; and Air Purifier, *WATERFIRST* = 0 with Water Purifier, *WATERFIRST* = 1.

⁸ Specifically, we bootstrap at the village level by randomly sampling villages from our original study sample. For each of 1000 bootstrapped samples of villages, we then repeat our analyses to obtain 1000 sets of results.



Fig. 3 Mean WTP for air and water purifier, USD per household per month—distribution of bootstrapped results **a** *WATERFIRST* unspecified, **b** *WATERFIRST* = 1 and **c** *WATERFIRST* = 0

6 Discussion

Prominent critiques of the contingent valuation method (e.g., Diamond and Hausman 1994) have focused on a distinct "order effect" in the results obtained from such surveys, whereby the stated WTP for an alternative is found to be higher when it is evaluated earlier in a sequence of multiple options. Others have identified an "embedding effect," in which stated WTP for a good varies depending on whether it is evaluated on its own or as part of a more inclusive category (Kahneman and Knetsch 1992). These concerns could be problematic for valuation of specific health technologies in low-income contexts, where households face multiple health risks. If, for instance, respondents are unable to determine which specific health risks reduction devices), stated WTP for the technology may be biased upward. At the same time, the existence of multiple health risks may impact WTP for specific health risk reductions directly. For example, if the burden of disease imposed by household air pollution is a relatively small proportion of the total household-level disease burden, then the WTP a for reduction in air-pollution-related risks will likely be lower than if it had comprised a

larger share of the total burden. This is because investments made to reduce specific health risks may do little to improve overall mortality or morbidity outcomes. Predictions about households' WTP for environmental health risk reductions using theoretical frameworks that consider multiple risks are therefore ambiguous (Jeuland et al. 2015d), and of limited use in guiding prioritization of different environmental and health policies.

Our unique study design allows us to consider these potential biases in contingent valuation surveys, and to investigate whether multiple disease interactions may influence households' preferences for disease-specific health-risk reductions. Our survey elicits private valuations for technologies that would reduce respiratory infection and diarrheal disease risks, the second and fourth leading contributors to the burden of disease globally (Murray and Lopez 2013), and two of the leading contributors to the global preventable environmental burden of disease. An overwhelming proportion of these illnesses affect people in low- and middle-income countries. Given that our sample is drawn from poor rural areas in Rajasthan, these risks are likely to be salient and distinct in our respondents' minds.

We find some evidence that the position of a specific technology in the sequence of two choices matters, but the significance of this effect is inconsistent across two different disease frames, which raises doubts about whether they arise from pure order effects. Indeed, as shown in Table 4, the order in which the two technologies are presented to respondents only appears to consistently matter for the water purifier. These results suggest that households' perceived private benefits from investments that mitigate diarrheal risks are greater than those that mitigate air pollution risks. Furthermore, the demand for the private water and air purifier is greatest when some of the health risk related to air pollution has previously been reduced by a community-level air quality improvement. This suggests that there may be important perceived complementarities in risk reductions when the less salient respiratory risk is reduced first at very low cost.⁹ While these results warrant further empirical testing, they do suggest that the community-level health prevention behavior.

Our results point to a conceptualization of households' decisions about investing in private health-improving technologies as grounded in an understanding of the broader disease environment and associated health complementarities. Specifically, the private benefits that accrue to households from investing in private environmental health technologies depend crucially on the full set of risks the household faces. For instance, a water filter can lead to significant improvement in health outcomes if waterborne diseases are the key health risk faced by the household. That same water filter, however, may make little difference in households' health outcomes if households also face health risks related to malaria, tuberculosis,

⁹ Because the disease frames were perfectly correlated with the order in which alternatives are presented to respondents, our interpretation of the results depends on the assumption that respondents took the information about the community-level intervention into account when answering the subsequent questions about the air and water purifier (though this is also of course a general critique of any stated preference design that assumes that respondents consider the information with which they are provided). As noted by a reviewer, respondents may have disregarded this information. In this specific instance, our results would not highlight the salience of complementarities in disease-risk reductions in respondents' minds. However, they would still provide informational value on (i) the relative importance households in rural India place on air- and water-based risk reductions; and (ii) an upper bound for their WTP for technologies that achieve these risk reductions on their own (because their budget constraints bind less). That said, we do not believe respondents ignored the information about the community-level intervention. That we observe the typical "order effect" result seen in the literature (whereby respondents are more likely to accept a price offer for alternatives presented earlier in the sequence) consistently only for the water purifier suggests that disease frame and, in turn, complementarities in disease-risk reductions matter. A more complete experiment would also have randomly varied the inclusion of the community-level risk reduction programs, but budget and sample size limitations prevented us from including the two additional arms that would have been required to test this formally.



Fig. 4 Health production function and decisions to invest in health technologies

and air pollution. Consequently, households' demand for the technology across these two situations may be very different.

More formally, this may be represented as in the S-shaped health production function shown in Fig. 4. Imagine that a community-level air quality intervention is perceived to improve the disease environment from a baseline of D_0 to D_{1A} , considerably more than that from D_0 to D_{1W} due to a community-level water quality intervention. Under these two very different reference points, additional investments in household-level water and air filters that further improve the disease environment-reducing each disease risk by equal amounts—may lead to very different improvements in health. Specifically, the acquisition of a water filter given a community-level air quality intervention further improves the disease environment from D_{1A} to D_{2W} while that of an air filter given a community-level air quality intervention improves it from D_{1W} to D_{2A} . The corresponding perceived increase in private health outcomes in these two scenarios is $(A_{2W} - A_{1A})$ and $(A_{2A} - A_{1W})$, respectively. As shown in Fig. 4, $(D_{2W} - D_{1A}) = (D_{2A} - D_{1W})$; however, $(A_{2W} - A_{1A}) > (A_{2A} - A_{1W})$. In other words, an equivalent improvement in the disease environment may be perceived to lead to a lower improvement in health outcomes depending on where one believes one is along the health production function. Assuming households' willingness to pay is proportional to perceived health benefits, willingness to pay for a technology that provides an equivalent improvement in the disease environment would be different in the two cases.

A rigorous exploration of why perceptions of multiple risks and preferences may be aligned in ways that leads to such perceived complementarities—especially given the fact that the burden of disease due to lower respiratory infections is estimated to be greater than that due to diarrheal diseases in low-income communities—is beyond the scope of this paper. It may be the case that individual episodes of diarrhea, which predominantly impacts young children, are identified more distinctly by caregivers and key decision-makers within the household. In contrast, adults also bear a relatively high burden of disease due respiratory ailments (Lozano et al. 2012). That an overwhelming majority of respondents in our sample indicate that both—respiratory infections and diarrheal diseases—are "equally dangerous" (73%) and "equally important to reduce" (87%) while 53% indicate that diarrhea is "worse" for children under five years old suggests that many perceive children to have greater relative sensitivity to diarrhea.

7 Conclusion

Households in low-income settings face a multitude of risks across different health domains. This can lead to low demand for health technologies that only reduce health risks in a single disease domain if they fail to sufficiently reduce overall mortality or morbidity rates. Using a unique contingent valuation survey design, we evaluate the extent to which health-risk reductions in one health domain impact willingness to pay for environmental health technologies in another domain. Specifically, we focus on respiratory infections and diarrhea-two diseases that are responsible for a large proportion of the global burden of disease, much of it disproportionately in low- and middle-income countries. We find that conditional on having already mitigated some of the health risks related to air pollution via a hypothesized community-level intervention, households are more (less) willing to make additional private health investments that mitigate risks related to diarrheal diseases. In contrast, households' willingness to pay for private technologies to reduce air-quality-related risks conditional on having mitigated community-level water-quality risks is considerably lower. Consistent with these results, we provide evidence on household-level perceptions that suggests that diarrheal diseases are a more salient source of health risks for households, and that perceived private benefits from diarrheal risk reductions outweigh those from comparable reductions in risks due to air pollution.

Our results have important implications for policymakers and practitioners working to improve environmental health outcomes in low-income settings. In particular, interventions that aim to promote technological solutions to reduce "domain-specific" health risks may fail to achieve widespread success in settings characterized by health risks in multiple domains. In these cases, implementers may be able to enhance effectiveness of interventions by combining solutions that address risks in multiple domains in potentially cost-effective ways (for instance, by providing bundles of complementary technologies that address distinct health risks).

It is also worth noting that the source of many of the health risks faced by the poor in lowincome settings is well beyond the confines of the household. The capacity of individuals to reduce these risks using household-level interventions is often constrained, such as in the case of ambient air pollution, which may require action at the community or regional level to see improvements. Optimal policies in these instances would need to recognize both household-level constraints and the potential for health-risk complementarities in innovative ways.¹⁰ We believe this presents a fruitful avenue for policy-relevant research going forward, especially in low-income settings.

¹⁰ For instance, a nominal fixed surcharge assessed as part of a household's water tariffs may, counterintuitively, make households more likely to connect to the water network (and reduce water-quality-related health risks) if proceeds from this surcharge are used to drive efforts to reduce complementary health risks in other domains. Utility payments, however, are not relevant in the context of rural Rajasthan, where a more straightforward payment vehicle-charging fees directly for the provision of specific risk-reducing technologies-may be most appropriate. We note, however, that at the community scale, even these simple payment approaches may require efforts to overcome collective action problems related to collective payments.

Appendix

A Survey Scenarios

The following pages, extracted from the English translation of our survey instrument, provide additional insight about the two scenarios—Option "A" and "B"—randomly allocated to survey respondents.

Option A. I want you to imagine that there was a program instituted in your community - at a small cost to you of 50 Rs. per year - which reduces the risk of respiratory illness in your community, because it increases the risk of respiratory illness. The device would have no other effects in your community, except 1 for reducing respiratory illness. For example, instead of 3 out of every 20 people in your community becoming ill with respiratory disease every 2 weeks, with this device, only 2 of every 20 people in your community would become ill with the disease every 2 weeks. (Enumerator: Please show card A1) Everyone in your community would be forced to contribute 50 Rs. to the purchase of this device by the village chief, there would be no exceptions This program would not affect diarrheal disease at all. But you could reduce diarrhea in your family by spending your own money on a water purification device. As shown in this picture, this device would decrease the 2-week diarrhea risk for each person in your family without changing anything else about the wat er, but it would cost you 300 / 1000 Rs. per year (Enumerator: Please show card A2) I want you to think carefully about whether you would purchase this device. When you think about this, you should consider your household budget and the other things that you could do with this money, if you did not spend it on the water purification device. Please answer the way you feel. H.5. Remember, you already paid 50 Rs. for a community air pollution program. Now I want to know: Would you pay 300 / 1000 Rs. per year for this water purification device? [-9] Don't know / not sure (Continue) [0] No (Continue) [1] Yes (Skip to H.7) H.6. Why would you not purchase this device? (Multiple answers possible) [1] It is too expensive / I have no money [4] My family risk is low [2] I have more important things to spend money [5] I do not believe the device will work [-9] Don't know / not sure [-95] Other, please specify: _ [3] Someone else should pay [6] I do not like the device H.7. How certain are you of your choice? [1] Very certain [2] Certain [3] Somewhat certain [4] Not certain [-9] Don't know / not sure H.8. Now I'd like you to imagine your decision to buy the water purification device for 300/1000 Rs. (Show card A2) without the community air pollution program, so you would not pay 50 Rs.and your respiratory risk would not change. Would you buy the water purifier, which will reduce your family's risk of dia mean if the community air pollution program is not institututed? [0]No [1]Yes [-9] Don't know / not sure H.9. Now I want to know how your decision in the previous question (H.8) would change if the compulsory community air pollution program was again present in your village (show respiratory risk reduction in card A1). How would your decision change? Would you be more or less likely to buy the water purifier, which reduces diarrhea (show card A2 again). for your family for 300/1000 Rs? [0] No change [1] More likely [2] Less likely Now I want you to make a second choice, which would further decrease your family's risk of respiratory disease. You could do this by spending your own money on an air purifier for your home that would remove contaminants from the air in your house. This device would allow you to decrease your respiratory r isk without changing anything else about the air in your house, but it would again cost you 300 / 1000 Rs. per year. (Enumerator: Please show card A3) H.10. Remember, you already paid 50 Rs. for a community air pollution program. Now I want to know: Would you pay 300 / 1000 Rs. per year for this air purification device? [0] No (Continue) [1] Yes (Skip to H.11) H.11. Why would you not purchase this device? [1] It is too expensive / I have no money [2] I have more important the set [-9] Don't know / not sure [4] My family risk is low [-9] Don't know / not sure [2] I have more important things to spend money on [3] Someone else should pay [6] I do not believe the device will work [6] I do not like the device [-95] Other, please specify: H.12. How certain are you of your choice? [2] Certain [3] Somewhat certain [4] Not certain [-9] Don't know / not sure [1] Very certain H.13. Now I'd like you to imagine your decision to buy the air purifier for 300/1000 Rs. (Show card A3) without the community air pollution program, so you would not pay 50 Rs.and your respiratory risk would not change. Would you buy the air purifier, which will reduce your family's risk of respiratory dis ease, if the community air pollution program is not institututed? [01No [-9] Don't know / not sure [1]Yes H.14. Now I want to know how your decision in the previous question (H.13) would change if the compulsory community air pollution program was again present in your village (show respiratory risk reduction in card A1). How would your decision change? Would you be more or less likely to buy the air purifier for your family, which reduces respiratory risk (show card A3 again), for 300/1000 Rs? [0] No change [1] More likely [2] Less likely (Skip to Section I) Option B. I want you to imagine that there was a program instituted in your community – at a small cost to you of 50 Rs. per year – which reduces the risk of diarrhea illness in your community. This program would involve installing a device at all community drinking water sources that would reduce the contami nants in the water you drink. This device would have no other effects in your community except for reducing diarrheal disease. For example, instead of 2 out of every 20 people in your community becoming ill with diarrheal disease every2 weeks, with this device, only 1 out of every 20 people in your community would become ill with this disease every 2 weeks. (Enumerator: Please show card B.1) Everyone in your community would be forced to contribute 50 Rs. to the purchase of this device by the village chief; there would be no exception s. This program would not affect respiratory illness at all. But you could reduce respiratory illness in your family by spending y our own money on an air purifier for your home that would remove contaminants from the air in your house. This device would allow you to decrease your respiratory risk without changing anything else about the air in your house, but it would cost you 300 / 1000 Rs. per year. (Enumerator: Please show card B.2) I want you to think carefully about whether you would purchase this device. When you think about this, you should consider your household budget and the other things that you could do with this money, if you did not spend it on the air purification device. Please answer the way you feel. H.15. Remember, you already paid 50 Rs. for a community water pollution program. Nowl want to know: Would you pay 300 / 1000 Rs. per year for this air purification device? [0] No (Continue) [1] Yes (Skip to H.17) [-9] Don't know / not sure H.16. Why would you not purchase this device? 1] It is too expensive / I have no money [4] My family risk is low -9] Don't know / not sure [2] I have more important things to spend money on [5] I do not believe the device will work [-95] Other, please specify: [3] Someone else should pay [6] I do not like the device H.17. How certain are you of your choice? [1] Very certain [2] Certain [3] Somewhat certain [4] Not certain [-9] Don't know / not sure H.18. Now I'd like you to imagine your decision to buy the air purifier for 300/1000 Rs (Show card B2) without the community water pollution program, so you would not pay compulsory 50 Rs. Would you buy the air purfiler, which will reduce your family's risk of respiratory disease, if the community water pollution program is not institututed? [0]No [1]Yes [-9] Don't know / not sure H.19. Now I want to know how your decision in the previous question (H.18) would change if the compulsory community water pollu tion program was again present in your village (show diarrheal risk reduction in card B1). How would your decision change? Would you be more or less likely to buy the air purifier, which reduces respiratory disease (show card B2 again), for your family for 300/1000 Rs?

[0] No change [1] More likely [2] Less likely

1096

References

- Anenberg SC, Balakrishnan K, Jetter J, Masera O, Mehta S, Moss J, Ramanathan V (2013) Cleaner cooking solutions to achieve health, climate, and economic cobenefits. Environ Sci Technol 47:3944–3952
- Ashraf N, Berry J, Shapiro JM (2010) Can higher prices stimulate product use? Evidence from a field experiment in Zambia. Am Econ Rev 100:2383–2413
- Banerjee AV, Duflo E (2007) The economic lives of the poor. J Econ Perspect 21:141-167
- Bennett D (2012) Does clean water make you dirty? Water supply and sanitation in the Philippines. J Hum Resour 47:146–173
- Bensch G, Grimm M, Peters J (2015) Why do households forego high returns from technology adoption? Evidence from improved cooking stoves in Burkina Faso. J Econ Behav Organ 116:187–205
- Bensch G, Peters J (2015) The intensive margin of technology adoption—Experimental evidence on improved cooking stoves in rural Senegal. J Health Econ 42:44–63
- Beyene AD, Koch SF (2013) Clean fuel-saving technology adoption in urban Ethiopia. Energy Econ 36:605– 613
- Bhojvaid V, Jeuland M, Kar A, Lewis JJ, Pattanayak SK, Ramanathan N, Ramanathan V, Rehman I (2014) How do people in rural India perceive improved stoves and clean fuel? Evidence from Uttar Pradesh and Uttarakhand. Int J Environ Res Public Health 11:1341–1358
- Blanton E, Ombeki S, Oluoch GO, Mwaki A, Wannemuehler K, Quick R (2010) Evaluation of the role of school children in the promotion of point-of-use water treatment and handwashing in schools and households-Nyanza Province, Western Kenya, 2007. Am J Trop Med Hyg 82:664–671
- Boisson S, Kiyombo M, Sthreshley L, Tumba S, Makambo J, Clasen T (2010) Field assessment of a novel household-based water filtration device: a randomised, placebo-controlled trial in the Democratic Republic of Congo. PLoS ONE 5:e12613
- Boyle KJ, Welsh MP, Bishop RC (1993) The role of question order and respondent experience in contingentvaluation studies. J Environ Econ Manag 25:S80–S99
- Braveman P (2006) Health disparities and health equity: concepts and measurement. Annu Rev Public Health 27:167–194
- Brooks N, Bhojvaid V, Jeuland M, Lewis J, Patange O, Pattanayak S (2016) How much do alternative cookstoves reduce biomass fuel use? Evidence from North India. Resour Energy Econ 43:153–171
- Brown J, Hamoudi A, Jeuland M, Turrini G (2016) "Seeing, believing, and behaving: Heterogeneous Effects of an Information Intervention on Household Behaviors to Improve Water Quality" (Submitted)
- Carson RT (2000) Contingent valuation: a user's guide. Environ Sci Technol 34:1413–1418
- Carson RT, Flores NE, Hanemann WM (1993) "On the Creation and Destruction of Public Goods: The Matter of Sequencing," CUDARE Working Paper 690
- Clark J, Friesen L (2008) The causes of order effects in contingent valuation surveys: an experimental investigation. J Environ Econ Manag 56:195–206
- Clasen T, Schmidt W-P, Rabie T, Roberts I, Cairncross S (2007) Interventions to improve water quality for preventing diarrhoea: systematic review and meta-analysis. BMJ 334:782–782
- Cook J, Jeuland M, Maskery B, Whittington D (2011) Giving stated preference respondents "Time to Think": results from four countries. Environ Resour Econ 51:473–496
- Diamond PA, Hausman JA (1994) Contingent valuation: is some number better than no number? J Econ Perspect 8:45–64
- Dow WH, Philipson TJ, Martin X S i (1999) Longevity complementarities under competing risks. Am Econ Rev 89:1358–1371
- Dupas P, Robinson J (2013) Why don't the poor save more? Evidence from health savings experiments. Am Econ Rev 103:1138–1171
- Fewtrell L, Kaufmann RB, Kay D, Enanoria W, Haller L, Colford JM (2005) Water, sanitation, and hygiene interventions to reduce diarrhoea in less developed countries: a systematic review and meta-analysis. Lancet Infect Diseases 5:42–52
- Freeman MC, Quick RE, Abbott DP, Ogutu P, Rheingans R (2009) Increasing equity of access to point-of-use water treatment products through social marketing and entrepreneurship: a case study in western Kenya. J Water Health 07:527
- Gasana J, Morin J, Ndikuyeze A, Kamoso P (2002) Impact of water supply and sanitation on diarrheal morbidity among young children in the socioeconomic and cultural context of Rwanda (Africa). Environ Res 90:76–88
- Greene WH (2011) Econometric analysis, 7th edn. Pearson, London
- Guerrant RL, DeBoer MD, Moore SR, Scharf RJ, Lima AAM (2012) The impoverished gut—a triple burden of diarrhoea, stunting and chronic disease. Nat Rev Gastroenterol Hepatol 10:220–229

- Haab TC, McConnell KE (2002) Valuing environmental and natural resources: the econometrics of non-market valuation. Edward Elgar Publishing, Cheltenham
- Halvorsen B (1996) Ordering effects in contingent valuation surveys. Environ Resou Econ 8:485-499
- Hamoudi A, Jeuland M, Lombardo S, Patil S, Pattanayak SK, Rai S (2012) The effect of water quality testing on household behavior: evidence from an experiment in rural India. Am J Trop Med Hyg 87:18–22
- Hanemann WM (1994) Valuing the environment through contingent valuation. J Econ Perspect 8:19–43
- Hensher D, Shore N, Train K (2005) Households' willingness to pay for water service attributes. Environ Resour Econ 32:509–531
- Hoehn JP, Loomis JB (1993) Substitution effects in the valuation of multiple environmental programs. J Environ Econ Manag 25:56–75
- Hunter PR (2009) Household water treatment in developing countries: comparing different intervention types using meta-regression. Environ Sci Technol 43:8991–8997
- Institute for Health Metrics and Evaluation (2013) GBD: Profile India. http://www.healthdata.org/sites/default/ files/files/country_profiles/GBD/ihme_gbd_country_report_india.pdf. Accessed 10 April 2016
- Jessoe K (2013) Improved source, improved quality? Demand for drinking water quality in rural India. J Environ Econ Manag 66:460–475
- Jeuland M, Bhojvaid V, Kar A, Lewis J, Patange O, Pattanayak SK, Ramanathan N, Rehman I, Soo JT, Ramanathan V (2015a) Preferences for improved cook stoves: evidence from rural villages in north India. Energy Econ 52:287–298
- Jeuland M, McClatchey M, Patil S, Poulos C, Pattanayak SK, Yang J-C (2015b) Do decentralized community treatment plants provide better water? Evidence from Andhra Pradesh.' Duke Environmental and Energy Economics Working Paper EE-15-01
- Jeuland M, Orgill J, Shaheed A, Revell G, Brown J (2015c) A matter of good taste: investigating preferences for in-house water treatment in peri-urban communities in Cambodia. Environ Dev Econ 21:291–317
- Jeuland M, Pattanayak SK, Bluffstone R (2015d) The economics of household air pollution. Annu Rev Resour Econ 7:81–108
- Kahneman D, Knetsch JL (1992) Valuing public goods: the purchase of moral satisfaction. J Environ Econ Manag 22:57–70
- Kartman B, Stålhammar N-O, Johannesson M (1996) Valuation of health changes with the contingent valuation method: a test of scope and question order effects. Health Econ 5:531–541
- Kremer M, Leino J, Miguel E, Zwane AP (2011) Spring cleaning: rural water impacts, valuation, and property rights institutions. Q J Econ 126:145–205
- Lewis JJ, Pattanayak SK (2012) Who adopts improved fuels and cookstoves? A systematic review. Environ Health Perspect 120:637–645
- Longo A, Hoyos D, Markandya A (2015) Sequence effects in the valuation of multiple environmental programs using the contingent valuation method. Land Econ 91:20–35
- Lozano R, Naghavi M, Foreman K, Lim S, Shibuya K, Aboyans V, Abraham J, Adair T, Aggarwal R, Ahn SY, AlMazroa MA, Alvarado M, Anderson HR, Anderson LM, Andrews KG, Atkinson C, Baddour LM, Barker-Collo S, Bartels DH, Bell ML, Benjamin EJ, Bennett D, Bhalla K, Bikbov B, Abdulhak AB, Birbeck G, Blyth F, Bolliger I, Boufous S, Bucello C, Burch M, Burney P, Carapetis J, Chen H, Chou D, Chugh SS, Coffeng LE, Colan SD, Colquhoun S, Colson KE, Condon J, Connor MD, Cooper LT, Corriere M, Cortinovis M, de Vaccaro KC, Couser W, Cowie BC, Criqui MH, Cross M, Dabhadkar KC, Dahodwala N, Leo DD, Degenhardt L, Delossantos A, Denenberg J, Jarlais DCD, Dharmaratne SD, Dorsey ER, Driscoll T, Duber H, Ebel B, Erwin PJ, Espindola P, Ezzati M, Feigin V, Flaxman AD, Forouzanfar MH, Fowkes FGR, Franklin R, Fransen M, Freeman MK, Gabriel SE, Gakidou E, Gaspari F, Gillum RF, Gonzalez-Medina D, Halasa YA, Haring D, Harrison JE, Havmoeller R, Hay RJ, Hoen B, Hotez PJ, Hoy D, Jacobsen KH, James SL, Jasrasaria R, Jayaraman S, Johns N, Karthikeyan G, Kassebaum N, Keren A, Khoo J-P, Knowlton LM, Kobusingye O, Koranteng A, Krishnamurthi R, Lipnick M, Lipshultz SE, Ohno SL, Mabweijano J, MacIntyre MF, Mallinger L, March L, Marks GB, Marks R, Matsumori A, Matzopoulos R, Mayosi BM, McAnulty JH, McDermott MM, McGrath J, Memish ZA, Mensah GA, Merriman TR, Michaud C, Miller M, Miller TR, Mock C, Mocumbi AO, Mokdad AA, Moran A, Mulholland K, Nair MN, Naldi L, Narayan KMV, Nasseri K, Norman P, O'Donnell M, Omer SB, Ortblad K, Osborne R, Ozgediz D, Pahari B, Pandian JD, Rivero AP, Padilla RP, Perez-Ruiz F, Perico N, Phillips D, Pierce K, Pope CA, Porrini E, Pourmalek F, Raju M, Ranganathan D, Rehm JT, Rein DB, Remuzzi G, Rivara FP, Roberts T, León FRD, Rosenfeld LC, Rushton L, Sacco RL, Salomon JA, Sampson U, Sanman E, Schwebel DC, Segui-Gomez M, Shepard DS, Singh D, Singleton J, Sliwa K, Smith E, Steer A, Taylor JA, Thomas B, Tleyjeh IM, Towbin JA, Truelsen T, Undurraga EA, Venketasubramanian N, Vijayakumar L, Vos T, Wagner GR, Wang M, Wang W, Watt K, Weinstock MA, Weintraub R, Wilkinson JD, Woolf AD, Wulf S, Yeh P-H, Yip P, Zabetian A, Zheng Z-J, Lopez AD, Murray CJ (2012) Global and

regional mortality from 235 causes of death for 20 age groups in 1990 and 2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet 380:2095–2128

- Lucas ME, Jeuland M, Deen J, Lazaro N, MacMahon M, Nyamete A, Barreto A, von Seidlein L, Cumbane A, Songane FF, Whittington D (2007) Private demand for cholera vaccines in Beira, Mozambique. Vaccine 25:2599–2609
- Luoto J, Najnin N, Mahmud M, Albert J, Islam MS, Luby S, Unicomb L, Levine DI (2011) What point-of-use water treatment products do consumers use? Evidence from a randomized controlled trial among the urban poor in Bangladesh. PLoS ONE 6:e26132
- Martín-Fernández J, del Cura-González MI, Rodríguez-Martínez G, Ariza-Cardiel G, Zamora J, Gómez-Gascón T, Polentinos-Castro E, Pérez-Rivas FJ, Domínguez-Bidagor J, Beamud-Lagos M, Tello-Bernabé ME, Conde-López JF, Aguado-Arroyo Ó, Sanz-Bayona MT, Gil-Lacruz AI (2013) Economic Valuation of health care services in public health systems: a study about willingness to pay (WTP) for nursing consultations. PLoS ONE 8:e62840
- McPhail AA (1993) The "five percent rule" for improved water service: can households afford more? World Dev 21:963–973
- Miller G, Mobarak AM (2015) Learning about new technologies through social networks: experimental evidence on nontraditional stoves in Bangladesh. Mark Sci 34:480–499
- Mobarak AM, Dwivedi P, Bailis R, Hildemann L, Miller G (2012) Low demand for nontraditional cookstove technologies. Proc Nat Acad Sci 109:10815–10820
- Murray CJ, Lopez AD (2013) Measuring the global burden of disease. N Engl J Med 369:448-457
- Null C, Kremer M, Miguel E, Hombrados JG Meeks R, Zwane AP (2012) Willingness to pay for cleaner water in less developed countries: systematic review of experimental evidence. 3ie Syst Rev 006
- Orgill J, Shaheed A, Brown J, Jeuland M (2013) Water quality perceptions and willingness to pay for clean water in peri-urban Cambodian communities. J Water Health 11:489
- Orgill-Meyer J, Jeuland M, Albert J, Cutler N (2018) Comparing contingent valuation and averting expenditure estimates of the costs of irregular water supply. Ecol Econ 146:250–264
- Ouedraogo B (2006) Household energy preferences for cooking in urban Ouagadougou, Burkina Faso. Energy Policy 34:3787–3795
- Pattanayak SK, Pfaff A (2009) Behavior, environment, and health in developing countries: evaluation and valuation. Annu Rev Resour Econ 1:183–217
- Poulos C, Yang J-C, Patil SR, Pattanayak SK, Wood S, Goodyear L, Gonzalez JM (2012) Consumer preferences for household water treatment products in Andhra Pradesh, India. Soc Sci Med 75:738–746
- Powe N, Bateman I (2003) Ordering effects in nested 'top-down' and 'bottom-up' contingent valuation designs. Ecol Econ 45:255–270
- Rehfuess EA, Puzzolo E, Stanistreet D, Pope D, Bruce NG (2013) Enablers and barriers to large-scale uptake of improved solid fuel stoves: a systematic review. Environ Health Perspect 122(2), 120–130
- Rosa G, Majorin F, Boisson S, Barstow C, Johnson M, Kirby M, Ngabo F, Thomas E, Clasen T (2014) Assessing the impact of water filters and improved cook stoves on drinking water quality and household air pollution: a randomised controlled trial in Rwanda. PLoS ONE 9:e91011
- Rufener S, Mäusezahl D, Mosler H-J, Weingartner R (2010) Quality of drinking-water at source and pointof-consumption - drinking cup as a high potential recontamination risk: a field study in Bolivia. J Health Popul Nutr 28
- Samples KC, Hollyer JR (1990) Contingent valuation of wildlife resources in the presence of substitutes and complement. In: Johnson RL, Johnson GV (eds) Economic valuation of natural resources: issues, theory, and applications. Westview Press, Boulder, pp 177–192
- Thurber MC, Warner C, Platt L, Slaski A, Gupta R, Miller G (2013) To promote adoption of household health technologies, think beyond health. Am J Public Health 103:1736–1740
- Usmani F, Steele J, Jeuland M (2017) Can economic incentives enhance adoption and use of a household energy technology? Evidence from a pilot study in Cambodia. Environ Res Lett 12:035009
- van der Kroon B, Brouwer R, van Beukering PJ (2014) The impact of the household decision environment on fuel choice behavior. Energy Econ 44:236–247
- Van Houtven GL, Pattanayak SK, Usmani F, Yang J-C (2017) What are households willing to pay for improved water access? Results from a meta-analysis. Ecol Econ 136:126–135
- Walker CLF, Perin J, Katz J, Tielsch JM, Black RE (2013) Diarrhea as a risk factor for acute lower respiratory tract infections among young children in low income settings. J Glob Health 3
- Whittington D (2002) Improving the performance of contingent valuation studies in developing countries. Environ Resour Econ 22:323–367
- Whittington D, Jeuland M, Barker K, Yuen Y (2012) Setting priorities, targeting subsidies among water, sanitation, and preventive health interventions in developing countries. World Dev 40:1546–1568
- World Health Organization (2011) Guidelines for drinking-water quality. World Health Organization, Geneva

World Health Organization (2014) Global Health Estimates for 2000–2012. http://www.who.int/healthinfo/ global_burden_disease/estimates/en/index2.html. Accessed 10 April 2016

World Health Organization (2015) India: WHO Statistical Profile. http://www.who.int/gho/countries/ind.pdf. Accessed 10 April 2016

Affiliations

Alexandra K. Shannon¹ · Faraz Usmani² · Subhrendu K. Pattanayak^{3,4,5} · Marc Jeuland^{3,5,6,7}

Alexandra K. Shannon ashannon@jhu.edu

Faraz Usmani faraz.usmani@duke.edu

Subhrendu K. Pattanayak subhrendu.pattanayak@duke.edu

- ¹ Department of Population, Family and Reproductive Health, Johns Hopkins Bloomberg School of Public Health, Johns Hopkins University, Baltimore, MD, USA
- ² University Program in Environmental Policy, Nicholas School of the Environment, Duke University, Durham, NC, USA
- ³ Sanford School of Public Policy, Duke University, Durham, NC, USA
- ⁴ Nicholas School of the Environoment, Duke University, Durham, NC, USA
- ⁵ Duke Global Health Institute, Duke University, Durham, NC, USA
- ⁶ Institute of Water Policy, Lee Kuan Yew School of Public Policy, National University of Singapore, Singapore, Singapore
- ⁷ RWI-Leibniz Institute, for Economic Research, Essen, Germany